Author’s response to reviews

Title: Management of patients with early stage lung cancer – why do some patients not receive treatment with curative intent?

Authors:

Ross Lawrenson (ross.lawrenson@waikatodhb.health.nz)
Chunhuan Lao (chunhuan.lao@waikato.ac.nz)
Leonie Brown (Leonie.Brown@waikatodhb.health.nz)
Lucia Moosa (Lucia.Moosa@healthshare.co.nz)
Lynne Chepulis (lynne.chepulis@waikato.ac.nz)
Rawiri Keenan (rawiri.keenan@waikato.ac.nz)
Jacquie Kidd (jacquie.kidd@aut.ac.nz)
Karen Middleton (Karen.Middleton@waikatodhb.health.nz)
Paul Conaglen (Paul.Conaglen@waikatodhb.health.nz)
Charles de Groot (Charles.deGroot@waikatodhb.health.nz)
Denise Aitken (Denise.Aitken@lakesdhb.govt.nz)
Janice Wong (Janice.Wong@waikatodhb.health.nz)

Version: 3 Date: 23 Jan 2020

Author’s response to reviews:

Editor Comments:

1. Please confirm in the section 'Ethics Approval and Consent to Participate' whether informed consent, written or verbal, was obtained from all participants and clearly state this in your manuscript. If verbal, please state the reason and whether the ethics committee approved this procedure. If the need for consent was waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this, including the name of the IRB or a reference to the relevant legislation.

Response: We have added “As this was a retrospective observational study using anonymised data, individual patient consent was not deemed a requirement by HDEC.”
2. In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response: We have added “While the overall study has been peer reviewed by the funding bodies, they have had no role in the design of the study and collection, analysis, and interpretation of data and or writing the manuscript.”

3. We notice that author Jacquie Kidd (JK) is missing from the authors’ contributions section. The individual contributions of all authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here:

http://www.biomedcentral.com/submissions/editorial-policies#authorship

Response: We have added “RL, CL, RK, CG, LC and JK were involved in the interpretation of data.”

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary files are cited within the text.

Response: We have uploaded a clean version.